Author’s response to reviews

Title: Empirical research in clinical supervision: A systematic review and suggestions for future studies

Authors:

Franziska Kühne (dr.franziska.kuehne@uni-potsdam.de)

Jana Maas (jamaas@uni-potsdam.de)

Sophia Wiesenthal (s_wiesenthal@posteo.de)

Florian Weck (fweck@uni-potsdam.de)

Version: 2 Date: 16 Jul 2019

Author’s response to reviews:

1. We note that abbreviations appear in the text but have not been included in the List of abbreviations. Please ensure that all abbreviations which appear in the text of the manuscript appear in the List of abbreviations section. If abbreviations are used in the text, they should be defined in the text at first use and included in this list.

>>> We have now added the missing abbreviations on p. 14.

2. Please move the Acknowledgements section so that it appears in the Declarations section.

>>> We moved the section.

3. When submitting your revised manuscript please ensure you do so as a single clean copy without any tracked changes, colored or highlighted text, as these are no longer required at this stage of the editorial process.

>>> We uploaded a clean copy of the manuscript.